Author's response to reviews

Title: Prevalence of peripheral arterial disease in patients at non-high cardiovascular risk. Rationale and design of the PANDORA study

Authors:

Claudio Cimminiello (claudio.cimminiello@avimercate.org)
Claudio Borghi (claudio.borghi@unibo.it)
Serge Kownator (s.kownator@wanadoo.fr)
Jean Claude Wautrecht (jwautrec@ulb.ac.be)
Christos P Carvounis (christos@ncmc.edu)
Stefanus E Kranendonk (skranendonk@tsz.nl)
Beat Kindler (bkindler@hin.ch)
Mario Mangrella (mario.mangrella@astrazeneca.com)

Version: 2 Date: 22 July 2010

Author's response to reviews: see over
Dear Dr Norton

Prevalence of peripheral arterial disease in subjects at moderate cardiovascular risk. Rationale and design of the PANDORA study (Manuscript ID 2022602249380845)

Claudio Cimminiello, Claudio Borghi, Serge Kownator, Jean Claude Wautrecht, Christos P. Carvounis, Stefanus Eliza Kranendonk, Beat Kindler, and Mario Mangrella, for the PANDORA Study Investigators

My co-authors and I, on behalf of the PANDORA Study Investigators, have made the following amends to our originally submitted manuscript.

- Listed all authors in the manuscript exactly as they are in the submission system, both in style and order.
- Included the name of the group within the list of authors.
- Ensured that table 1 is formatted as one complete table.
- Formatted the abbreviations into a paragraph.
- Formatted the authors’ contributions into a paragraph.
- Removed the visible vertical lines from the tables.
- Checked the manuscript for any typographical errors.

Thank you very much in advance.

My kindest regards.

Claudio Cimminiello